Author's response to reviews

Title: An ecological study of regional variation in work injuries among young workers

Authors:
F Curtis Breslin (cbreslin@iwh.on.ca)
Peter Smith (psmith@iwh.on.ca)
James R Dunn (jim.dunn@utoronto.ca)

Version: 7 Date: 14 May 2007

Author's response to reviews:

Monday May 14, 2007

Dr Annabel Phillips
Senior Assistant Editor
BMC-series Journals
BioMed Central

Dear Dr. Phillips:

Please find attached the revised manuscript entitled "An ecological study of regional variation in work injuries among young workers" by Dr. Curtis Breslin, Peter Smith, and James R. Dunn.

We have made the following changes to the paper:
1. We have ensured that the abstract in both the manuscript and submission system are identical.
2. I would prefer my name as author on the manuscript as is. All my published articles are under the name "F Curtis Breslin" and maintaining consistency would allow people to locate this article more easily. Please contact me if that is a problem.
3. All headings have been aligned to the left.
4. The title "List of abbreviations" is now "Abbreviations".
5. There is a note under the "Authors' contributions" section that all authors have read and approved the final manuscript.
6. A section called "Additional files" has been added at the end of the manuscript after the tables.
7. All additional files have been mentioned in the text in a numerically consecutive order.

If you require any additional information, please feel free to contact me by email, telephone, fax, or regular mail (contact information below). Or you may contact my assistant, Sudipa Bhattacharyya at extension 2159.

Many thanks for your time and consideration.

Sincerely,

F. Curtis Breslin, Ph.D.
Scientist
Institute for Work & Health
481 University Avenue, Suite 800
Toronto, Ontario
Canada M5G 2E9
Tel: (416) 927-2027 ext. 2225
Fax: (416) 927-4167
Email: cbreslin@iwh.on.ca